Author's response to reviews

Title: The Generation and Gender Shifts in Medicine: An Exploratory Survey of Internal Medicine Physicians

Authors:

Emily Jovic (ejovic@uwo.ca)
Jean E Wallace (jwallace@ucalgary.ca)
Jane Lemaire (lemaire@ucalgary.ca)

Version: 6 Date: 4 May 2006

Author's response to reviews:

Formatting changes requested
-------------------------------
Please reduce the length of your Abstract. We generally prefer abstracts to be of 350 words or less.

Figure and table placement text - Please remove the figure and table placement text. Figures and tables will be included in the final version of the manuscript as near as possible to their first mention in the manuscript.

We have made both sets of revisions accordingly.